Author’s response to reviews

Title: The analysis of relationship between selected sociodemographic factors and disorders of speech organs in Parkinson's patients

Authors:
Wioletta Pawlukowska (wsna@o2.pl)
Karolina Skonieczna-Żydecka (karzyd@sci.pum.edu.pl)
Iwona Rotter (iwrot@wp.pl)
Krystyna Honczarenko (khonczar@sci.pum.edu.pl)
Przemysław Nowacki (przemyslaw.nowacki@pum.edu.pl)

Version: 3 Date: 01 Dec 2017

Author’s response to reviews:

Responses to Editor Comments

Thank you very much for the comments to the article. We made the following changes in the article:

1. Title Page section, line 2, page 1- we have indicated, with an asterisk and annotation, the author for correspondence - according to the editor's recommendations

2. Abstract section page 2- We divided the summary into sections:

   - Background (the context and purpose of the study)

   - Methods (how the study was performed and applied tests)

   - Results (the main findings)

   - Conclusions (brief summary and potential implications);
3. Abstract section  line 1 page 2- We added a sentence to Background section

4. Background section  line 11 page 3 - we have changed the name “Parkinson’s disease” to abbreviation “PD”

5. Methods section page 4- We changed 'Materials and Method' to 'Methods

6. We have added a list of abbreviations page 9

7. We've added information about (page 9-10):

Declarations

- Ethics approval and consent to participate
- Consent to publish
- Availability of data and materials
- Competing interests
- Funding
- Authors' Contributions
- Acknowledgements

All the changes were made in accordance with the editors' guidelines